# 16. Politics of judicial governance

David Kosař and Katarína Šipulová<sup>1</sup>

The power of courts has increased worldwide at an unprecedented pace. At the same time. judicial governance has changed as well. Most importantly, several regions have witnessed a steady rise in judicial self-governance. While in 1985 only around 10 per cent of jurisdictions in the world had judicial councils or judicial appointment commissions, in 2015 these bodies participated in the selection of judges in almost 60 per cent of countries (Garoupa and Ginsburg 2015).

This phenomenon is truly global. Many common law countries introduced judicial appointment commissions, which eventually became a dominant model in the Commonwealth (BIICL 2015), spanning from Australia (Bunjevac 2020) to South Africa (Oxtoby 2021: Brett 2022) and England and Wales (Gee, Hanzell & Malleson 2015). Even within the United States several states implemented the so-called 'merit plan' (or 'Missouri plan'), which resulted in the rise of merit commissions involved in the selection of state judges (Volcansek 2009; Goelzhauser 2018). Several African and Asian countries entrenched a judicial service commission (e.g. Kenya, South Africa and Malaysia) or a judicial council (e.g. Tunisia and Bangladesh) in their constitutions (Oxtoby 2021; Bari 2022). Judicial councils spread also in Latin America, where they started to compete with the supreme courts over influence within the judiciary (Hammergren 2002; Bill Chavez 2005; Pozas-Loyo & Ríos-Figueroa 2018). Europe has gone even further. Many countries, such as France, Italy, Portugal, Spain, the Netherlands, and most recently Ireland have introduced judicial councils voluntarily (Kosař 2018; Castillo-Ortiz 2019). Virtually all post-communist states in Central and Eastern Europe did so under pressure from the European Union during the accession process (Bobek and Kosař 2014; Kosař 2016; Parau 2018).

The key element of all these reforms was the transfer of powers concerning judicial governance from political branches to judges and bringing in the expert element. The new judicial self-governance bodies decide primarily on issues concerning the careers of individual judges. Judicial councils usually have broader powers, spanning from decisions concerning the selection, promotion, and disciplining of judges to various housekeeping functions (Garoupa & Ginsburg 2015; Kosař 2016). Judicial appointment commissions have a narrower mandate as they decide merely on the selection of judges.

However, selection, promotion, and disciplining of judges and other decisions concerning the careers of individual judges are just a snapshot of judicial governance, which has undergone important developments in other areas, too. Judicial training has professionalised, and new specialised judicial academies have been introduced in many countries. Digitalisation, hastened by the COVID-19 pandemic, brought with it new tools and software. Even administrative decisions on the courts' functioning, such as the overall number of judges assigned to

a court, the number and composition of panels at each court, the overall number of assigned administrative personnel and law clerks, case allocation, and judicial performance evaluation, were overhauled. Moreover, there is an increasing institutional variety in exercising administrative governance within the judiciary, as these tasks can be implemented not only by traditional bodies such as judicial councils, the Supreme Court, court presidents, and the 'US-style' judicial conference complemented by the Director of the Administrative Office of the U.S. Courts, but also by the novel agencies such as the Court Service (Geyh 2021) and the 'Israeli-style' Director of Courts (Lurie et al. 2019; Bunjevac 2020).

This development reflects the growing demands placed on a modern client-oriented judiciary in the twenty-first century. The judiciary must be flexible and respond to novel challenges such as the COVID-19 pandemic, digitalisation, and calls for greater diversity. At the same time, old challenges have not disappeared. Politicians are still willing and able to tinker with the judiciary and align their decision-making with their preferences. Think of Hugo Chavez's frontal attacks on the Venezuelan judiciary (Taylor 2014), Recep Erdoğan's abrupt changes in judicial governance and purges within the Turkish judiciary after the failed coup d'état (Özbudun 2015; Esen & Gumuscu 2016; Varol, Pellegrina & Garoupa 2017; Olcay 2017), the Modi government's interference with judicial appointments in India (Khaitan 2020), or Benjamin Netanjahu's recent judicial reform proposal curtailing Supreme Court's constitutional review competence and imposing executive control over judicial appointments (Weiler 2023). Even within the European Union we can see a backlash against the rise of judicial self-governance as several judicial councils in Central and Eastern Europe were hollowed out (Jakab 2020) or captured (Śledzińska-Simon 2018; see also Chapter 33 by Petra Bárd, Nóra Chronowski and Zoltán Fleck in this volume).

These examples of a straightforward backlash against judicial self-governance by populist or authoritarian regimes show that the increasing involvement of judges in judicial governance is not a linear development. However, even in consolidated democracies some politicians as well as scholars have criticised the rise of judicial self-governance. They usually argue that the judiciary lacks democratic legitimacy, that too much judicial self-governance may lead to self-replication, non-responsiveness and corporativism of judges, and, more pragmatically, that judges do not have the necessary political capital to negotiate with other ministries the budgetary issues nor the political leverage to push through the necessary legislation in the parliament. These concerns resulted in including civil society members and non-lawyers in judicial appointment commissions (Gee et al. 2015), reducing the number of judges on a judicial council (Vauchez 2018), as well as retaining certain powers within the Ministry of Justice (Vasek 2022). This pushback against judicial self-governance took place in good faith and the relevant changes were made incrementally. Therefore, it should be distinguished from the backlash exercised by populist or authoritarian regimes.

no. 101002660).

<sup>&</sup>lt;sup>1</sup> The research leading to this project has received funding from the European Research Council (ERC) under the European Union's Horizon 2020 research and innovation program (INFINITY, grant

At the same time, political interferences with judicial governance triggered the proliferation of various international standards on the global<sup>2</sup> as well as regional<sup>3</sup> level. These international standards were initially treated as soft law, but their normative weight has increased over time. This development has also contributed to the judicialisation of judicial governance on the domestic as well as supranational level. On the domestic level, constitutional tribunals and supreme courts started reviewing judicial appointments, the disciplining of judges, and case assignment more thoroughly. On the supranational level, regional human rights courts, especially the Inter-American Court of Human Rights and the European Court of Human Rights, have been keen to shape domestic judicial design by creative interpretation of their founding documents (Kosař & Lixinski 2015). The European Court of Justice jumped on the bandwagon as well and developed a whole new set of requirements for judicial governance in order to respond to the attacks against the rule of law and judicial independence in Hungary, Poland. and Romania (Kochenov & Pech 2021; Kosař & Vincze 2022; Bustos 2022).

The aim of this chapter is to conceptualise judicial governance, address the main challenges (both old and new) it faces, and identify the new trends therein. In doing so we bring insights from law as well as political science. We also look beyond the formal rules and institutional templates and emphasise the politics of judicial governance and the role of informal institutions. Our major argument is three-fold. First, we argue that we must go beyond the executive and judicial councils and also study other actors of judicial governance such as chief justices and judicial associations. Second, we show that channels of politicisation of the judiciary never disappear completely. The creation of a judicial self-governance body does not make the power disappear or the dangers evaporate. Power is just transferred to other hands and new channels of politicisation of the judiciary are created (Spáč, Šipulová and Urbániková 2018; Spáč 2020). Third, informal institutions and gender norms are crucial for understanding judicial governance.

This chapter proceeds as follows. Section 1 conceptualises judicial governance and identifies its dimensions. Section 2 zeroes in on the growing number of relevant actors in judicial governance. Section 3 analyses its changing channels of politicisation. Section 4 identifies three recent trends in judicial governance (judicialisation, internationalisation, and standardisation) and their repercussions. Section 5 discusses informal aspects of judicial governance, which are often overlooked, yet form a proverbial glue that allows the smooth administration of the judiciary. Section 6 then demonstrates the importance of understanding the gender aspects of judicial governance. Section 7 concludes.

## JUDICIAL GOVERNANCE AND ITS DIMENSIONS

Judicial governance has been the buzzword for more than two decades. Questions on how best to balance principles of judicial independence and accountability, how to distribute the power between judges and politicians, insulate courts from political interference, prevent court-packing and telephone justice, and how to discourage judicial corruption and clientelism gradually increased in importance and became a salient topic of judicial studies scholarship as well as judicial reforms (on the clash between judicial independence and accountability see also Chapter 15 by Vanberg, Broman and Ritter in this volume). It is becoming evident that even the writ-small mechanisms such as panel composition, case allocation, and the internal flow of case files matter (Leloup & Kosař 2022).

Yet, the term judicial governance itself is often misunderstood and wrongly simplified to decisions on the selection and promotion of judges, or their disciplining and removal (Malleson & Russell 2006; Lee 2011; Bobek 2015; Castillo-Ortiz 2019). These issues are important, but judicial governance is a much broader field that concerns every single aspect of courts' functioning, including efficiency, transparency, ethical issues, and a more mundane day-to-day agenda of court administration, as well as more structural issues concerning the relationship of the judiciary with the executive and the legislature.

In order to plausibly capture and understand the politics of judicial governance, this chapter therefore opts for a broad holistic understanding of judicial governance developed in the latest scholarship (Börzel & Risse 2010; Kosař 2018; Castillo-Ortiz 2019; Bunjevac 2020) that defines it as 'a structured model of social coordination which produces and implements a set of institutions, rules, and practices which are collectively binding and which regulate how the judicial branch exercises its functions' (Šipulová et al. 2022). Judicial self-governance then captures the extent to which judges and courts participate in judicial governance.

While most scholarship on judicial governance, quite understandably, focuses on personal aspects concerning the careers of individual judges, such as the selection, promotion, and disciplining of judges, the concept of judicial governance is much broader. In order to see the developments within judicial governance more clearly, it is thus helpful to unpack judicial governance into smaller dimensions, each of them raising a specific set of issues and undergoing potentially different development (Kosař 2018; Bunjevac 2020).

Tentatively, there are eight such distinctive dimensions: regulatory, personal, administrative, financial, educational, informational, digital, and ethical (Kosař 2018; Bunjevac 2020; Šipulová et al. 2022). These eight dimensions are visualised in Table 16.1. Each of them aggregates a set of individual competences related to judicial governance.

So far, the most comprehensive list of such competences has been introduced in 'the Judicial Self-Governance Index', an analytical tool measuring the participation of judges in individual dimensions of judicial governance, irrespective of the institutional design of the field (Šipulová et al. 2022). The Judicial Self-Governance Index relied mostly on competences previously addressed or reflected by qualitative and quantitative scholarship4 deriving the competences from existing literature on judicial governance (Kosař 2018), governance of

See e.g. Arts. 9 and 13 of the 2010 Magna Carta of Judges (https://rm.coe.int/2010-ccje-magna -carta-anglais/168063e431), Arts. 2.3 and 3 of the 1999 Universal Charter of the Judge (https://www .iaj-uim.org/universal-charter-of-the-judge-2017/), Art. 32 of the 2010 Report on the Independence of the Judicial System (https://www.venice.coe.int/webforms/documents/default.aspx?pdffile=CDL -AD(2010)004-e), Bangalore principles of Judicial Conduct of 2006 https://www.judicialintegritygroup .org/images/resources/documents/ECOSOC 2006\_23\_Engl.pdf.

European Network of Councils for the Judiciary (2017). Performance Indicators 2017. Available at: https://www.encj.eu/images/stories/pdf/workinggroups/independence/encj\_report\_ia\_ga\_adopted \_ga\_13\_6.pdf.

Smithey and Ishiyama's index (2002) for example mentions regulatory dimension, Hayo & Voigt (2016) indexed the selection, nomination, approval, and dismissal of judges. Gutmann & Voigt (2018) correlated the transfer of judges and cases, and Feld & Voigt (2003) operationalised powers related to the transparency and publication of case law as part of judicial governance. Similarly, budgetary arrange-

Table 16.1 Dimensions of judicial governance

I. Regulatory	Competences related to establishment, abolition, or changes in the jurisdiction and procedural rules of a court
II. Administrative	Composition of a court (setting the number of judges, panels, and their composition), work schedules, case assignment
III. Personal	Selection and (re)appointment of judges, promotions, removals, and transfers of judges (permanent and temporary), disciplining of judges, civil and criminal prosecution, evaluations of judges
IV. Financial	Size of a court's budget, salaries of judges
V. Educational	Compulsory education (plan and structure) and further training and education of judges
VI. Informational	Publication of rulings, recordings of trials, annual reports, case assignment, disclosure of judges' property, political affiliation, and some personal information
VII. Digital	Placement of servers with online data
VIII. Ethical	Preparation and interpretation of the code of conduct, extrajudicial activities of judges, communication with media

Source: Šipulová et al. 2022.

agencies (Verhoest 2013; Lurie et al. 2020; Mathieu et al. 2017), judicial independence, and effectiveness, as well as data collected by CEPEJ<sup>5</sup> and EU Justice Scoreboard.<sup>6</sup>

The regulatory dimension relates to the entrenchment of courts and judicial systems in the constitution or statutory law. Due to their regulatory character, the powers belonging to this dimension (establishment or abolition of courts, changes in jurisdiction and courts' structure. statutes and legal procedural rules) are wielded mainly by legislative power (parliament). With subsequent judicial empowerment, however, we have witnessed increasing, although formally subtle, engagement of courts, courts presidents, and judicial councils. Once established. judicial councils (or potentially chief justices) can be consulted on any systemic legislative changes in the regulation (Kosař and Šipulová 2018). Although their positions are typically not binding, they offer judicial actors' bodies an important channel for stepping inside the regulatory framework and utilising informal powers and networks to influence this dimension of judicial politics.

The administrative dimension targets the seemingly mundane, day-to-day functioning of courts: decisions on the number of judges assigned to a court, the structure of single- and multi-judge panels, as well as their respective compositions. Administrative powers also include the number and quality of administrative personnel or clerks, oversight of the system of case assignment (and reassignment), or evaluation of courts' overall performance (quality of decisions, backlog, public spending). While the administrative dimension might seem less salient than selection and removal processes, it actually significantly impacts both the effectiveness of decision-making, as well as judicial independence itself. Many political or third-party interferences, particularly in democratising regimes, attempt to utilise administrative powers to shift the balance at courts and exert pressure on individual panels or judges.

ments, determination of judges' salaries, promotions, evaluations, and management of courts' tasks were included in older indices of judicial independence (Van Dijk 2021).

Case allocation is particularly sensitive administrative issue as some players might be tempted to attempt to assign their case to a friendly judge (Kosař 2016; Marcondes et al. 2019).

The personal dimension naturally attracts the most political and scholarly interest and lies at the very heart of judicial governance. This dimension covers all decisions on selection, the careers and removal of judges, including decisions on their accountability via disciplinary (or criminal) motions. The dimension also covers the evaluation of judicial performance and ties it to decisions on financial bonuses and similar measures. The personal dimension is the most contested one, as the transfer of power to select judges from political branches of power to the judiciary itself has to fulfil both pragmatic (insulation from political pressure) as well as theoretical and doctrinal (democratic legitimacy) tests of justification.

The financial dimension is much narrower and covers the financial or budgetary administration of courts: decisions on the size and allocation of a court's budget and judges' salaries. Financial competences are held almost exclusively by political actors, yet the distribution of power in this dimension is slowly attracting more attention and raising controversies in many, at least European, jurisdictions.

The educational dimension captures decisions on the compulsory education and training of judicial candidates and judges. In recent years, we have seen a significant transfer of power from the executive branch to judges and independent agencies (such as judicial academies and associations) that took over many of the educational competencies previously carried out by ministries of justice (Wittreck 2018; Levi-Faur 2009; Lurie et al. 2019). There is a growing scholarship on judicial training (Dallara & Piana 2016), but the educational dimension has usually not been connected to the broader phenomenon of judicial governance (but cf. Parau 2018; Fagan 2019; Wittreck 2006; Benvenuti and Paris 2018). Yet, actors wielding educational powers both decide on the structure and content of these systems of education and significantly impact the pool of potential candidates eligible for the office of judge, as well as create expectations on the quality and scope of knowledge judges should have.

The informational dimension concerns the relationship between judges (courts) and the public. Competences in the informational dimension set out decisions on the extent of transparency and visibility of judicial decision-making (publication and communication of cases, annual reports, and statistics) but also on the personal affairs of individual judges (disclosures of property, party affiliations, etc.).

The digital dimension is a rather young aspect of judicial governance. It results from the growing corpus of data and databases available at courts. For instance, the body that regulates the where the servers with the case-law and internal court documents are located has a wide-reaching impact both on the internal management of courts, and on the de facto degree of transparency courts can actually achieve (e.g. ability to manage their own clouds and servers, or the opportunity to create new search engines). The digital dimension can thus also contribute to the visibility and accessibility of information on courts.

Finally, the ethical dimension is very closely related to various disciplinary mechanisms against judicial misbehaviour. However, it is typically less formal, vested in the hands of a different actor (such as an ethical committee), and takes into account a different set of considerations than the traditional disciplinary measures. This dimension concerns, in particular, decisions on the preparation and interpretation of the code of judicial conduct, communication of judges with media or public (Ginsburg and Garoupa 2009), and the regulation of their extrajudicial activities.

European Commission for the Efficiency of Justice, regular evaluation of European judiciaries available at https://www.coe.int/en/web/cepej.

An EU tool, part of the Rule of Law Toolbox, available at https://ec.europa.eu/info/policies/justice -and-fundamental-rights/upholding-rule-law/eu-justice-scoreboard en.

Each of these eight dimensions of judicial governance has a different degree of political salience and a different relationship with judicial independence, accountability, diversity, efficiency, and legitimacy. The transfer of some of these powers from the political branches to judges or 'fourth branch' institutions (and vice versa) can therefore be driven by different considerations and goals. For instance, the delegation of personal competences to judges pursues the goal of the insulation of judges from political pressure, but it still requires some extent of political engagement to give judges legitimacy and prevent corporativism and judicial corruption. Administrative competences are also predominantly held by judges, but in this dimension it is often a pragmatic decision that results from judges' greater expertise in the day-to-day functioning of the judiciary, the informational gap between ministries of justice and court presidents, and the need for greater responsiveness from administrative governance to the actual needs of courts.

In sum, it is necessary to study each dimension of judicial governance separately and only then to make claims about judicial governance as a whole, since it is quite possible that within the same country several dimensions may undergo different development. Each dimension of governance might be organised around different interests, and individual competences can be distributed among different sets of actors (see Section 2). For instance, decisions on the establishment, abolition, merger, division, and jurisdiction of courts are typically under the control of the legislature, even though judges have increasingly gained the ability to comment on and sometimes even shape judicial reforms via judicial councils or other bodies, in which they have the majority. Similarly, administrative decisions on the courts' functioning, such as the overall number of judges assigned to a court, the number and composition of panels at each court, the overall number of administrative personnel and law clerks, case allocation. and judicial performance evaluation, can be exercised by a variety of actors - the executive. the legislature, a judicial council, the Court Service, court presidents, the 'US-style' judicial conference complemented by the Director of the Administrative Office of the US Courts, or the 'Israeli-style' Director of Courts. In other words, it is necessary to understand the variety of actors in judicial governance and their relationship, to which we turn next.

### ACTORS OF JUDICIAL GOVERNANCE: BEYOND THE EXECUTIVE AND JUDICIAL COUNCILS

The politics of judicial governance is often narrated through the 'judges vs politicians' lenses. The whole debate on the ideal distribution of power is thus framed via the question, Which branch of state power should have more powers in (a given dimension of) judicial governance?. However, this false dichotomy prevents us from understanding the complexity of actors, networks, and interests affecting judicial governance, and offers only a limited picture of how much impact judges actually have on judicial governance.

The eight-dimensional structure of judicial governance includes a constellation of actors and institutions, typically represented by state bodies, judges, lawyers, politicians, and what we call 'judicial self-governance bodies': institutions established to take part in individual judicial governance competences, including at least one judge (Kosař 2018). These are typically judicial councils, court services, judicial appointment commissions, the Supreme Court, the chief justice, court presidents, judicial associations, and judicial academies.

Understandably, judicial councils have attracted most attention recently, because they are most visible and epitomise the judicial empowerment movement. They are also heavily promoted by supranational institutions that have considered them as the best bulwark against political interferences with the judiciary and an institutional guarantee of judicial independence. The burgeoning scholarship on judicial councils, which offers various categorisations of their strength and powers, showed though that their contribution to judicial independence or the quality of democracy is less clear and certainly not linear in all cases (Garoupa & Ginsburg 2015; Castillo-Ortiz 2019). Others show that their success is based on contingent circumstances such as embedded norm of professionalism in the Brazilian judiciary (Pozas-Loyo & Ríos-Figueroa 2023).

However, it is also necessary to stress that judicial councils still do not exist in many countries, and even in those where they do, they offer only a fragment from the whole picture of how judicial governance works and how individual competences are organised among multiple actors. Although we can observe a certain convergence of supranational recommendations towards the strong judicial council dominated by judges (see Section 4), the models of judicial councils established across the world are actually quite diverse. They differ in composition (ratio of judges, politicians, and experts), powers (the number of judicial governance dimensions they are involved in), as well as in actors with whom they share these powers. The same claim applies to judicial appointment commissions, prevalent in the common law world, as their rationale and design vary a lot from one country to another (BIICL 2015; Bunjevac 2020; Brett 2022).

In fact, the mere existence of a judicial council or a judicial appointment commission does not tell us much about the participation of judges in the judicial governance, since in all countries representatives of the ministry of justice, court presidents, the Supreme Court, judicial associations, politicians, and/or prosecutors participate in judicial governance to a certain degree as well. This became even clearer during the COVID-19 pandemic, when the ministers of justice used emergency powers to curtail court operation and shape judicial governance more broadly (Lurie 2021). The influence of judges on judicial appointments is not static either. Modi's and Netanjahu's judicial reforms show that politicians want to regain their powers and shape judicial appointments without the major input of judges (Khaitan 2020; Weiler 2023).

The most recent scholarly works have also documented the rise of smaller actors such as judicial academies, directors of courts, and chief justices (Verhoest 2013; Lurie et al. 2020; Kosař & Spáč 2021). Judicial networks, which operate on both domestic and transnational level, became important actors of judicial governance (Dallara & Piana 2016), who are sometimes criticised for the lack of democratic legitimacy (Parau 2018). Combined with the creation of new areas of regulation, judicial governance is becoming a significantly decentralised field with a high level of power distribution.

Interestingly, even in the countries where political branches still have the major say in judicial governance, such as Austria, Czechia, and Germany, judges can play a significant role. In Austria that is so because the key positions within the ministry of justice are actually filled by judges who are temporarily assigned to the ministry (Vasek 2022). In Czechia, it results from the fact that the ministry of justice informally delegated significant powers to court presidents who, due to their expertise, are better equipped to supply the short-term needs of judiciaries (Blisa et al. 2018). Contrary to general wisdom, Germany also shows a significant dose of judicial self-governance, since it features as many as eight judicial self-governance bodies.

Germany just advances a different conception of judicial self-governance (than a judicial council model) which reflects the prevailing German understanding of democratic legitimacy and separation of powers (Wittreck 2018).

This means that judicial empowerment is not a phenomenon exclusive to the establishment of judicial councils, but may permeate all institutional constellations of judicial governance (Šipulová et al. 2022). Vice versa, the creation of a strong judicial council dominated by judges does not prevent the further engagement of political actors in judicial governance as the executive may still decide on the court budgets, regulate the internal functioning of the court, appoint court presidents, or take part in the selection or promotion of judges.

In sum, three interim conclusions can be made. First, the model of judicial governance does not in itself tell us who controls a given dimension of judicial governance. Second, the division of competences between politicians and judges is never absolute. Instead, both politicians and judges have a say in judicial governance. Judicial self-governance is thus a matter of degree and operates on the continuum rather than in the 'either-or' fashion. Third, judicial empowerment is not necessarily linear, as many countries have recently witnessed pushbacks against it (Uitz 2015).

These findings also suggest that the binary 'judges vs politicians' logic, employed by the dominant judicial governance scholarship (Parau 2018; Castillo-Ortiz 2019; Mikuli et al. 2019), is flawed because it ignores other actors of judicial governance that do not come from any of these three branches of power. Very recent scholarship has observed a new trend of 'agencification' (Lurie et al. 2019). The gradual growth in the powers of many judicial governance actors has been accompanied by the increasing autonomy of their position vis-à-vis the judiciary, the legislature, as well as the executive (Jordana & Sancho 2004; Mathieu et al. 2017). This is very true even for some judicial councils and the perception of their role by other judicial governance actors. In the end, the majority of judicial councils are of mixed composition, opening up a vexing question which branch of power individual members represent. or to what extent they execute their offices as completely independent agencies. Compared to supranational recommendations, which clearly identify judicial councils as judicial bodies, the question to be pursued by theoretical scholarship is what position individual actors of judicial governance have within the system of separation of powers (Kadlec, Šipulová & Kosař 2022).

#### 3. CHANNELS OF POLITICISATION: OLD AND NEW

The ideological alignment of judges, especially at supreme and constitutional courts, is an important benefit for every government. To make it happen, the executive and the legislature in the past often used their influence over the sword or the purse<sup>7</sup> to shape judicial governance. The Ministry of Justice, the Presidential administration, and the monarchy, each in its own way, found channels for politicising the judiciary. Politicisation of the judiciary reached its apex in the communist countries where the omnipotent Party carefully screened new judges, kept the elected judges on a short leash by short renewable terms and oversight by the General Prosecutor, dismissed or persecuted judges who dared to stand in the way of the socialist legality, gave instructions to judges on how to decide politically salient cases (a practice colloquially referred to as 'telephone justice'), and assigned those salient cases to reliable hard-core communist judges in order to achieve the 'right' outcome (Kühn 2011; Ledeneva 2008).

In consolidated democracies, many channels of politicisation of the judiciary have closed or have been exposed to public scrutiny. In the United States, the selection of Article III federal judges has remained deeply political, many aspects of judicial governance have been depoliticised, and the decisions thereon transferred to the Judicial Conference of the United States and the circuit judicial councils. In most Commonwealth countries, judicial appointment commissions took charge of many judicial governance issues. A similar trend of growing judicial self-governance took place in Europe and Latin America.

However, the creation of a judicial self-governance body does not make the power disappear or the dangers of politicisation evaporate. Power is just transferred to other hands and new channels of politicisation of the judiciary are created. These channels differ from one jurisdiction to another. European experience is particularly insightful in this regard. The Slovak judiciary was politicised through the dominant role of the Chief Justice in the judicial council (Spáč, Šipulová and Urbániková 2018). The Polish judiciary has recently been politicised not only by the Minister of Justice, but also through court presidents and the new members of the National Council of the Judiciary elected by the parliamentary majority (Śledzińska-Simon 2018). In France and Italy, the major channels of politicisation of the judiciary are not the non-judicial members of their judicial councils, but judicial associations (Guarnieri 2004; Benvenuti and Paris 2018; Vauchez 2018). A recent scandal in Italy showed that politicians used judicial associations as a proxy for protecting their interests (Sallusti & Palamara 2021). In Germany, the main channel of politicisation is the promotion committees (Wittreck 2018). In Hungary, the major channel of politicisation of the judiciary is the new National Office for the Judiciary that took the key powers away from the Hungarian judicial council (Uitz 2015). In Spain and Turkey, politicisation of the judiciary has flourished due to the selection of judicial members of the judicial council by political branches. The difference is that while the Spanish judicial council has been captured by political parties (Torres 2018), in Turkey it is the presidential administration that currently has the major grip over the judicial council (Calı & Durmus 2018).

Outside Europe, politicisation of judicial governance came from both the political branches and the judiciary. While Hugo Chavez used virtually all means to get the Venezuelan judiciary under control (Taylor 2014), in Mexico it was the Supreme Court judges who created patronage networks that maintained their grip over the judiciary (Pozas-Loyo & Ríos-Figueroa 2018). In Georgia, judicial selection has been dominated by judicial oligarchies using judicial councils to channel their power and influence (Tsereteli 2020). In China, the communist party controls the courts via party committees, party meetings, and training, opening the floor to growing judicial corruption (Wang & Liu 2021). In Zimbabwe, president Mugabe employed several techniques aimed to control the judiciary, from packing the Supreme Court to the removal of judges who refused to resign (Castagnola 2018). In Senegal, Uganda, and Madagascar, attempts at judicial review of election results led to episodes of violence (Llanos 2015), assassination (The Judiciary Insider 2018), or seizures of judges' property (Llanos 2015).

In sum, the Ministry of Justice model of judicial governance is increasingly viewed as an anachronism, a remnant of the past that should be replaced by an autonomous self-regulated and depoliticised judiciary (Mikuli et al. 2019). However, lessons from across the globe tell us that judicial councils and other judicial self-governance bodies do not necessarily close the channels of politicisation of the judiciary. Judicial councils can be captured not only from the

Hamilton in Federalist No. 78.

outside (Popova 2010; Torres 2018), but also from the inside (Pozas-Loyo & Rios-Figueroa 2018; Spáč, Šipulová and Urbániková 2018). Unfortunately, the Polish scenario also attests that politicians always find some judges who are willing to cooperate with them, no matter how obvious the sinister intentions of the judicial reform are (Śledzińska-Simon 2019).

### STANDARDISATION, JUDICIALISATION, AND INTERNATIONALISATION OF JUDICIAL GOVERNANCE

There are three major trends in judicial governance that go hand in hand: standardisation. judicialisation, and internationalisation. The standardisation encompasses various efforts to unify certain aspects of judicial governance and turn them into recommendations and later on into universal or at least regional standards. At the universal level it is difficult to find consensus and thus there has been little progress since the United Nations Bangalore Principles of Judicial Conduct (2002), despite the efforts of the United Nations Special Rapporteur on the independence of judges and lawyers. Other organisations and associations try to fill this gap. Mt. Scopus International Standards of Judicial Independence (2008) are probably the most advanced.

At the regional level, there has been more development recently, especially in Europe The Venice Commission, the European Commissions as well as the advisory bodies such as the Consultative Council of European Judges and the European Network of Councils for the Judiciary have produced dozens of opinions, declarations, and reports on most aspects of judicial governance (Bobek & Kosař 2014; Kosař 2016; Parau 2018; ENCJ 2021, De Visser 2015). Even more recommendations, guidelines, standards exist on the domestic level.

Once the standards are at place, courts have the benchmarks for reviewing the legislative and executive acts concerning the judiciary. This in turn reinforces judicialisation of judicial politics. Of course, constitutional tribunals and supreme courts reviewed judicial reforms that interfered with judicial interference even before the supranational standards emerged. However, they usually focused on few selected issues such as disciplining, impeachment and removal of judges. That is no longer true, and we can see an increasing judicialisation of other areas of judicial governance across the globe. The Canadian Supreme Court declared the appointment of its new member, Marco Nadon, to be unconstitutional (Mathen 2015). The Indian Supreme Court struck down the constitutional amendment that changed the system of selection of supreme courts judges by transferring this power from the collegium of supreme court judges to the National Judicial Appointments Commission (Sengupta and Sharma 2018). The Italian Constitutional Court formulated the basic principles of constitutional conformity for process of cutting salaries of judges.8 The Spanish Supreme Court abolished the salary bonuses, the German Federal Administrative Court allowed the judicial review of case assignment. 10 In Central and Eastern Europe as well as in Latin America virtually every judicial reform ends up before constitutional courts too (Kosař 2017; Sadurski 2019; Helmke 2017).

Judicialisation is further reinforced by internationalisation of judicial governance. In the Global South, the World Bank and the International Monetary Fund have been particularly active in shaping domestic judicial governance by their rule of law and judicial independence initiatives (White 2009). More recently, regional human rights courts, especially the European Court of Human Rights and the Inter-American Court of Human Rights, have been increasingly moving beyond their original mandates, and making determinations about the design of national courts and their governance, encouraging domestic judicial reforms (Kosař & Lixinski 2015; Leloup & Kosař 2022). In the European Union, the level of internationalisation and judicialisation of judicial governance reached a whole new level after the European Court of Justice abandoned its deference to Member States in this area and delivered the landmark Portuguese Judges judgment (Bonelli & Claes 2018). Since then, the European Court of Justice developed a massive case law that set several requirements for judicial governance in new as well as old European Union Member States (Leloup 2020; Kochenov & Pech 2021, Moraru & Bercea 2022; Kadlec & Kosař 2022). The European Court of Human Rights tries to catch up and these two supranational courts now thus engage in intensive cross-fertilisation of judicial governance ideas that sometimes go too far (Leloup & Kosař 2022; Karlsson 2022).

Of course, not all countries are witnessing all of these three developments, at least not to the same degree. While European Union contributed heavily to all three trends, the United States have been resistant to all of them. Other regions show that supranational pressure may work well even without judicialisation. A typical example is Southern Africa, where several countries replaced opaque informal appointment systems inherited from the colonial era by merit-based system with judicial appointment commissions (Brett 2022). Brett shows that the rise of the merit orthodoxy in this region does not result primarily from judicialisation, but rather from the mix of domestic and supranational pressures that reflect broader social development in the decolonisation context. Finally, these three developments do not get through uncontested and are not irreversible. In fact, even within the European Union there is a considerable backlash against some of these trends, for instance in Kaczinski's Poland and Orbán's Hungary (Sledzinska-Simon 2018; Sadurski 2019; Uitz 2015). These two leaders want to dejudicialise politics in general and the politics of judicial governance in particular (Petrov 2022).

What is important to note is, however, that all three trends largely ignore the complexity of judicial governance as a field. Repeated political tinkering with courts' composition and independence, and increasing democratic backsliding encouraged standardisation of judge-dominated judicial governance and the vigorous judicial protection of this judicial design, but it overlooks the negative empirical experience of post-communist, post-authoritarian and developing countries (Hammergren 2002; White 2009; Kosař 2018; Šipulová et al. 2022). The international standards, now backed by supranational courts, perceive judicial governance as best organised by judges, ideally in a judicial council. This understanding is based on a conflation of judicial interests with interests in independent, fair and rule of law governed

See Judgment of the Italian Constitutional Court sp. No. 223/2012 of 11.10.2012.

See Judgment of the Spanish Supreme Court no. 2004\30 (STS, 3a) of 7. 3. 2006. See also Contini. Francesco & Mohr, Richard. Reconciling Independence and Accountability in Judicial Systems. Utrecht Law Review. 2007, vol. 3, no. 2, pp. 34-35.

See the judgment of German Federal Administrative Court of 28. 11. 1975 (BVerwGE 50, 11 = NJW 1976, 1224).

The nature and effects of this European transnational judicial dialogue is further discussed by Law in Chapter 17 of this volume.

judiciary. Very few supranational bodies recognise and reflect threats of corporativism and judicial corruption, since they mostly rely on judges and self-governance as a bulwark against political interferences.

#### 5. JUDICIAL GOVERNANCE AND INFORMALITY

It is generally accepted that there is a great deal of informality in politics, but this wisdom is often forgotten when it comes to judicial politics. Informal exercise of power politics in judicial governance is perhaps even more important, as the decisions behind the closed doors in this area of governance may have significant repercussions for the rule of law (Zgut 2022). It is thus particularly important to discern what 'the proverbial room where it happens' is, 12 who sits at its table, and what informal rules those sitting at the table apply.

When discussing the engagement of actors, transfers of power, or politicisation channels in judicial governance, we often noted de facto powers or the ability of various players to utilise their informal influence. From the conceptual point of view, there are three standard ways in which the scholarship engages with informality and informal institutions in judicial governance: (1) through the prism of constitutional conventions; (2) from the institutional perspective which focuses on informal rules and practices; and (3) from a relational perspective that studies informal networks.

Constitutional conventions are typically explored by legal scholarship (Stephenson 2021; Sirota 2011). Although they are not framed as capturing the informal dimension of judicial governance, 13 they do entail a large portion of informality and rely on deeply rooted and repeated practices and rules that do not have a clear bearing in the written law. The majority of constitutional conventions related to the area of judicial governance revolve around the selection and appointment of apex courts' judges or chief justices (Melton and Ginsburg 2014), or questions of judicial independence (Grove 2018). In Israel, the President appoints judges 'in accordance with the selection of the Committee for the Selection of Judges.' The unwritten convention is that the President is in fact bound by the opinion of the Committee and cannot deviate from the Committee's list. Similar practices have been recently confirmed by the European Court of Human Rights at the backdrop of Icelandic system of appointment of judges (Karlsson 2022). In Germany, a constitutional convention concerning the election of Federal Constitutional Court judges allocates each of the major political parties a seat on the Bench to nominate an occupant on (Taylor G. 2014; Kischel 2015). The Supreme Court of Canada recognised a constitutional convention related to remuneration of judges as part of the rule of law and judicial independence guarantees.14

In sum, constitutional conventions are typically unwritten, yet socially followed and perceived as binding. They fill the gaps in written law and sometimes even get 'absorbed by law' (Sirota 2011), if recognised as binding by domestic courts. For example, in 2020 the Supreme Court of Israel acknowledged the enforceable character of the constitutional convention according to which the Knesset appoints one governmental and one opposition member for

12 Hamilton musical. 13 Technically, conventions may include both formal and informal institutions. Therefore, they

See Remuneration of Judges of the Provincial Court (PEI) [1997] 3 SCR 3 (SCC).

cannot easily be categorised as a subgroup of informal institutions (Stephenson 2021; Sirota 2011).

the two parliamentary seats in the Committee for the Selection of Judges (Lurie 2022). The major puzzle related to conventions is whether they are more fragile and vulnerable to arbitrary change or expropriation and can easily lead to swift constitutional decay (Issacharoff & Morrison 2020), or to the contrary, whether they are so deeply embedded and socially shared that they can resist the attempts to change them through new legislation.

On the other hand, informal institutions are a domain of social science research. They are often described as the invisible social glue of political systems (Jakab 2020; Dunoff & Pollack 2018), filling in the gaps of formal regulation. They are created outside officially sanctioned channels (Helmke & Levitsky 2004). Their interaction with formal rules and practices is quite complex: they can complement, accommodate, but also compete with or even replace formal institutions (Helmke & Levitsky 2006). They emerge either where formal institutions exist but are incomplete, ineffective, too difficult to change, or contradictory to actors' (publicly non-acceptable) goal (Helmke & Levitsky 2005; Lauth 2015), or in the space where formal institutions do not exist at all (Lauth 2015).

Informal institutions are essential for judicial independence and the rule of law. Depending on their consonance with values underlying formal institutions, they can either subvert or protect the rule of law and the quality of democracy. The dissonance between formal and informal rules and practices is sometimes described as the hollowing out of democratic institutions. For example, 'gentlemen's agreements' between judicial associations may compete with or even substitute for formal rules governing the selection and promotion of judges (Pierson 2000; Pozas-Loyo & Rios-Figueroa 2018). These pacts may in turn entrench patronage (Guarnieri 2013; Benvenuti & Paris 2018; Vauchez 2018), nepotism (Spáč 2020) and vertical gender segregation (Sofos 2020). Similarly, politically savvy chief justices can tweak the formal rules and forge informal alliances with politicians (Kosař & Spáč 2021), with other court presidents (Kosař 2017), or with transnational judicial networks (Dallara & Piana 2015; Parau 2018). Informal practices like corruption (Popova 2012b), telephone justice (Popova 2012a; Ledeneva 2008), and clientelism (Popova & Beers 2020) may undermine existing formal institutions.

On the other hand, many informal institutions also have positive effects. Well-functioning informal institutions may increase the efficiency and quality of judicial decision-making, and, in the long run, also increase the resilience of formal democratic institutions. Interestingly, compared to legal research which perceives conventions as too susceptible to revision, social scientists argue that informal institutions are more difficult to change (than formal frameworks) because they are deeply embedded in social behaviour and less transparent to individual actors (North 1991). Judges typically take part in various informal networks, learning best practices across supranational levels (Dothan 2021). Overall, however, the informal rules and practices with neutral or positive effects are much less explored, with only a few pioneering studies engaging with judicial associations and transnational networks, norms diffusion, and inter-court dialogues.

Yet, the workings of informal institutions, particularly in European jurisdictions, are heavily under-studied. The existing scholarship so far has focused mostly on negative repercussions of informality in Latin America (Pozas-Loyo & Ríos-Figueroa 2018) and South-East Asia (Dressel, Urribarri & Stroh 2017; Harper & Colliou 2022) and the detrimental effects of corruption, nepotism, and patronage on selection processes and judicial independence. Only a few studies have explored the role of judicial culture in democratic decay in European countries (Jakab 2020; Zgut 2022), or the role of informal networks in selection processes in the USA

(Bird & McGee 2022). The largest number of studies on informality engaged with probability and patterns of judicial decision-making (Randazzo 2008).

Overall, a little more attention, although largely incidental, has been paid to informal networks created among actors of judicial governance. From the relational perspective (Dressel, Urribarri & Stroh 2017), it is important to acknowledge that judges are embedded in various circles of social interactions and their behaviour (on and off bench) is shaped by relational flows in networks to which they belong. This observation is mostly ignored by legal scholarship as well as the supranational approach to judicial governance, that prefers to see judges as independent, autonomous on other actors, and unburdened by any polarising interests apart from the delivery of justice. As studies from the post-communist countries however show. judicial networks and networks judges take part in significantly shape clientelistic relations inside the judiciary, and manage to deform the results of formal designs of merit-based selection processes (Tsereteli 2022).

Although the relational perspective allows us to see also many positive informal institutions, leading to stronger judicial dialogues, social responsiveness, or the legitimacy of courts, it is the networks interfering in judicial independence that attract more academic interests, Yet, uncovering extra-judicial networks (connections of judges to politicians and third actors) is extremely difficult. While Tünde Hando's proximity to Viktor Orbán is well known (Uitz 2015), to uncover such informal relations in other jurisdictions might be extremely difficult, yet crucial. For instance, in Slovenia one can hardly assess the functioning of the judicial council without knowing about the dense web of informal networks that made important decisions outside the judicial council (Avbelj 2018). In France, Italy, and Spain it is crucial to know who belongs to which judicial association (Vauchez 2018; Benvenuti & Paris 2018; Torres Pérez 2018). In Czechia court presidents have created several informal groups that have a major say in key areas of judicial governance (Blisa et al. 2018). Informal networks may also affect different stages of the recruitment of judges, in both Europe (Spáč 2018) and the Americas. The rules and practices created within these networks can completely replace existing formal arrangements. For example, gentlemen's agreements between judicial associations more or less replaced formal rules on the selection and promotion of judges in Mexico (Pierson 2000; Pozas-Loyo & Ríos-Figueroa 2018), and significantly deform selection processes in Italy (Benvenuti 2018). In the post-communist area, politically savvy chief justices still manage to tweak the formal rules and forge informal alliances with politicians (Kosař & Spáč 2021; Tsereteli 2022), with other court presidents (Kosař 2017), or with transnational judicial networks (Dallara & Piana 2015; Parau 2018) In China, institutional proximity between the Party, the administrative apparatus, and the courts facilitated judicial corruption (Wang & Liu 2022).

Fortunately, recent political science scholarship has made significant progress in conceptualising and analysing such informal networks (Dressel, Urribarri & Stroh 2017, 2018), and it is high time to apply these insights in legal scholarship and, even more importantly, in practice. As we have demonstrated above, supranational bodies, particularly two European courts, are increasingly active in shaping the regulation and policies of judicial governance. Yet their monitoring of institutional systems remains blind to the informal sphere of politics (Zgut 2022). As we have demonstrated in this section, informal institutions are difficult to capture and change; however, they are crucial for understanding how the judiciary works in practice and they play an indispensable role in the effective and efficient functioning of formal democratic frameworks.

## JUDICIAL GOVERNANCE AND GENDER DIVERSITY

Judicial governance can serve many goals. Constitutional and supranational courts tend to emphasise judicial independence and the rule of law.15 However, new public management expects courts also to become accountable, transparent, efficient, and quality-oriented. Judicial governance thus should deliver, and in fact balance, these often-competing values (Mak 2008; Dunoff & Pollack 2017). To make things even more complicated, there is a growing consensus that courts should attend to the challenges of equality and diversity (Malleson 2009; Resnik 2021). Hence, judicial governance should be designed to promote not only the rule of law and new public management values, but also diversity of the judiciary (Malleson 2009; Grossman 2012; Rackley 2013). Gender diversity has gradually become the most prominent, albeit not the only one (Resnik 2021; Weinshall 2022), issue in diversifying the judiciary.

It is worth noting that the idea of (gender) diversity on the bench is generally accepted irrespective of its eventual impact on courts' decision-making, since the evidence on whether female and male judges decide cases differently is still conflicting (Boyd at al 2010: 392; Peresie 2005; Tate & Handberg 1991; Songer and Johnson 2007; Weinshall-Margel 2011; Eisenberg et al. 2012). The arguments for gender diversity typically include positive effects on public trust in the judiciary (Resnik & Dilg 2006), structural impartiality of courts (Lawrence 2010; Chen 2003), a better diversity of experience and knowledge (Weinshall 2022; Resnik 2021), following by increased quality due to the enlarged pool of candidates (Rackley 2013: 25-27).

However, until recently, gender aspects of judicial governance have been underresearched and most studies on female judges focused primarily on descriptive gender representation and barriers for access of women to judicial profession (Arana et al. 2021; Arrington et al. 2021). This research explains women's access to courts with different structural and institutional factors that are often interrelated. Female judges benefit from (1) improved educational possibilities in law for women increase the pool of eligible women judges (Williams and Thames 2008; Sonnevold 2017; Sonnevold & Lindbekk 2020), (2) changes in cultural gender norms towards leadership and family life (Duarte et al. 2014; Harwa 2016), and (3) recruitment of judges based on transparency, objective merit-based criteria, and formal rules rather than on discretion, opaqueness, and informal patronage networks that tends to benefit men (Schultz & Shaw 2013; Kenney 2013; Boigeol 2013). In other words, introducing more merit-based and transparent appointment procedures for judges based on competitive examinations has often helped women circumvent the largely male power networks that previously excluded them from the judiciary (Tøraasen 2022).

The problem of access of women to the judiciary permeates most common law countries. Civil law countries fare better in terms of overall gender representation in courts. However, if women have the same chances to enter the judiciary, it does not necessarily mean that they will progress like their male colleagues. Several studies actually show that in judiciaries with majority of female judges women still face 'glass ceilings' and struggle to reach the apex courts (Valdini & Shortell 2016; Goldar 2020). The barriers to progress are similar to barriers of access, namely (1) opaque and informal process of promotion of judges (Zheng et al. 2017, Pozas-Loyo & Ríos-Figueroa 2018; Escobar-Lemmon et al. 2021) and (2) gender

<sup>15</sup> See above.

norms resulting in different work-life balance of women and men (Schultz 2013; Kalem 2020; Havelková et al. 2021). A similar pattern applies to other important positions within the judiciary such as the chief justices and court presidents that tend to be dominated by men (Havelková et al. 2021), even though Africa shows that a considerable progress is possible even this area (Dawuni & Kang 2015). Nevertheless, women still face more obstacles if they want to reach positions of power and influence within the judiciary.

So far only few studies analysed the impact of introducing expert bodies on gender renresentation. Existing studies concern mostly common law countries and judicial appointment commissions or merit commissions (Iyer 2013; Blackwell 2017; Dawuni & Masengu 2019-Masengu 2019; Escobar-Lemmon et al. 2021). They tell a cautionary tale. While replacing the executive models of judicial governance by judicial councils and judicial appointment commission might professionalise the selection and promotion of judges, it may do so slowly and incrementally (Iyer 2013) or only if other conditions are met (Malleson 2006). Moreover, expert bodies do not necessarily eradicate privilege and power dynamics since they may create a different type of dynamic that can be harmful for women (Masengu 2019). The existing research also shows that there is no one size fits all solution for consolidated, developing and post-conflict societies with widely diverging general gender norms. Counterintuitively, in some countries 'gender-neutral' judicial reforms aimed at strengthening the judiciary or the bureaucratisation of the judiciary have done more for women's judicial representation than explicitly gender-targeted policies that often meet stiff resistance (Jasper 2022; Tøraasen 2022). Finally, to our knowledge, the role of gendered norms in other areas of judicial governance beyond the selection and promotion of judges such as case assignment, composition of panels, judicial training, and extrajudicial activities of judges has not been studied thoroughly at all. Future research should explore these areas systematically as well.

#### 7. CONCLUSION

The development of dynamics in judicial governance have mirrored the rise, pushback, and backlash against judicialisation politics and the increasing importance of the courts. In a few decades we have seen a shift from executive-led judicial governance models to judicial councils and other judge-dominated bodies (judicial self-governance) and, more recently, attempts to dilute judicial power in the governance and administration of courts by including civil society members and other non-partisan actors and to construct judicial councils as more autonomous agencies standing beyond all three state powers.

Compared to international optimism accompanying the boom of judicial councils, recent empirical studies suggest that reliance on judge-dominated judicial governance is very problematic (Bobek and Kosař 2014; Bobek 2015, Kosař 2018; Spáč 2020; Kosař and Spáč 2021; Šipulová et al. 2022), that it does not bring with it more efficiency or judicial independence (Gutmann & Voigt 2018; Hayo & Voigt 2016), nor does it offer better protection from political interferences (Varol et al. 2017). Strikingly though, in particular European supranational policies seem to continue ignoring these findings.

In this chapter we have provided a bird's eye view of the key policies and most contested issues of judicial governance. First, the judicial governance field is broad and should not be conflated with the selection and disciplining of judges. As we have demonstrated, it has dozens of areas organised in various dimensions whose importance is gradually increasing. Second, it

is a multi-actor field. Recent trends demonstrate that we need to look beyond judicial councils as, even in governance models with judicial councils, several other actors, from ministries of justice to judicial academies, retain significant powers.

Third, the dynamics of judicial governance or the rise of judicial councils cannot anymore be explained solely through the binary judges vs politicians logic. Empirical experience from several countries suggests that (a) judges hold significant powers in ministerial as well as judicial council models, they are nested inside various bodies with mixed composition, and none of these actors operates in a vacuum - instead, they cooperate and share judicial governance powers; (b) studies from non-European regions suggest that the proliferation of judicial councils was, in fact, motivated not by power distribution within the three branches, but by social and supranational pressure (Brett 2022; Garoupa 2022).

Fourth, we noted two trends, agencification and power de-concentration, in the field of judicial governance, which suggest that individual judicial governance actors can no longer been squeezed into the three traditional state powers. Instead, they increase their autonomy oradually become a guarantor institution rather than body that represents any of the three traditional branches.

These considerations are important for a proper understanding of the power dynamics in the judicial governance field, particularly in the face of increasing challenges to judicial councils based on pragmatic (willingness of politicians to capture and control the courts) and normative (lack of legitimacy of courts to govern) considerations.

As we have demonstrated in this chapter, judicial governance is a highly complex phenomenon the contours of which go far beyond the selected model of court administration, since even judges in the Ministry of Justice model of court administration can have significant powers. The number of actors and agencies that participate in judicial governance has gradually increased, and has brought more expertise and less partisanship into the field (Kosař and Spáč 2021; Kosař and Blisa 2018). Accordingly, the focus of scholarship on judicial governance and politics should be redirected from judicial councils to other actors.

At the same time, judicial councils require more theorising. While if well designed they can eliminate some political interferences, they are also known to freeze informal rules and practices present within the judiciary. This brings us to the need to reconceptualise judicial councils at the backdrop of new literature on the fourth branch institutions and autonomous agencies (Tushnet 2021; Khaitan 2022). More attention should be paid to the perceptions and expectations of judicial councils in respect of interests they should represent (as a part of the judiciary, a coordination body between representatives of all three branches, or a post-branch institution that is completely autonomous on any of classical three powers, Kadlec, Šipulová and Kosař 2022).

### REFERENCES

Arana Aráya, Ignacio, Melanie M. Hughes & Aníbal Pérez-Liñán. (2021). 'Judicial Reshuffles and Women Justices in Latin America,' American Journal of Political Science 65: 373-88.

Arrington, Nancy B., Leeann Bass, Adam Glynn, Jeffrey K. Staton, Brian Delgado & Staffan I. Lindberg. (2021). 'Constitutional Reform and the Gender Diversification of Peak Courts,' American Political Science Review 115: 851-68.

Avbelj, Matej. (2018). 'Contextual Analysis of Judicial Governance in Slovenia,' German Law Journal 19: 1901-30.

- Bari, M. Ehteshamul. The Functioning of the Supreme Judicial Council, the Changes Introduced to the Method of Removal of Judges of the Supreme Court of Bangladesh in 2014 and the Subsequent Scathing Attack on the Judiciary. In: The Independence of the Judiciary in Bangladesh (Springer, 2021) 137–55.
- Benvenuti, Simone & Paris, Davide. (2018). 'Judicial Self-Government in Italy: Merits, Limits and the Reality of an Export Model,' German Law Journal 19: 1641–70.
- Benvenuti, Simone. (2018). 'The Politics of Judicial Accountability in Italy: Shifting the Balance, European Constitutional Law Review 14: 369–93.
- BIICL. The Commonwealth. The Appointment, Tenure and Removal of Judges under Commonwealth Principles: A Compendium and Analysis of Best Practice (London, 2015).
- Bill Chavez, Rebecca. (2005). 'The Appointment and Removal Process for Judges in Argentina: The Role of Judicial Councils and Impeachment Juries in Promoting Judicial Independence,' Latin American Politics & Society 49: 33–58.
- Bird, Christine & McGee, Zachary. (2022). 'Going Nuclear: Federalist Society Affiliated Judicial Nominees' Prospects and New Era of Confirmation Politics,' American Politics Research 1–20.
- Blackwell, Michael. (2017). 'Starting Out on a Judicial Career: Gender Diversity and the Appointment of Recorders, Circuit Judges, and Deputy High Court Judges, 1996–2016,' Journal of Law and Society 44: 586–619.
- Blisa, Adam, Papoušková, Tereza & Urbániková, Marína. (2018). 'Judicial Self-Government in Czechia: Europe's Black Sheep?,' German Law Journal 19: 1951–76.
- Bobek, Michal and Kosar, David. (2014). 'Global Solutions, Local Damages: A Critical Study in Judicial Councils in Central and Eastern Europe,' German Law Journal 15: 1257–92.
- Bobek, Michal. Selecting Europe's Judges: A Critical Review of the Appointment Procedures to the European Courts (Oxford University Press, 2015).
- Boigeol, Anne. 'Feminisation of the French "Magistrature" in Ulrike Schultz and Gisela Shaw (eds) Gender and Judging (2013) 125–42.
- Bonelli, Matteo & Monica Claes. (2018). 'Judicial serendipity: How Portuguese judges came to the rescue of the Polish judiciary,' EuConst 14: 622–43.
- Börzel, Tanja & Risse, T. (2010). 'Governance without a state: Can it work?,' Regulation and Governance 4: 113-34.
- Boyd, Christina L., Lee Epstein & Andrew D. Martin. (2010). 'Untangling the causal effects of sex on judging,' American Journal of Political Science 54: 389–411.
- Brett, Peter, 'The new politics of judicial appointments in Southern Africa,' (2022 forthcoming) Law and Social Inquiry, 1–31.
- Bunjevac, Tom. Judicial Self-Governance in the New Millennium (Springer, 2020).
- Bustos Gisbert, Rafael. (2022). 'Judicial Independence in European Constitutional Law,' European Constitutional Law Review 18: 591–620.
- Çalı, Başak & Durmuş, Betül. (2018). 'Select Judicial Self-Government as Experimental Constitutional Politics: The Case of Turkey,' German Law Journal 19: 1671–1706.
- Castagnola, Andrea. Manipulating Courts in New Democracies: Forcing Judges off the Bench in Argentina (Routledge, 2018).
- Castillo-Ortiz, Pablo. (2019). 'The politics of implementation of the judicial council model in Europe,' European Political Science Review 11: 503–20.
- Chen, Edward M. (2003). 'The Judiciary, Diversity, and Justice For All. Asian Law Journal,' 10: 127–42. Dallara, Cristina & Daniela Piana. Networking the Rule of Law: How Change Agents Reshape Judicial Governance in the EU (Routledge, 2016).
- Dawuni, Josephine J., & Alice J. Kang. (2015). 'Her Ladyship Chief Justice: The Rise of Female Leaders in the Judiciary in Africa,' Africa Today 62:2 45–69.
- Dawuni, Josephine J. & Tabeth Masengu. 'Judicial Service Commissions and the Appointment of Women to Higher Courts in Nigeria and Zambia' in Susan M. Sterett & Lee Demetrius Walker (eds) Research Handbook on Law and Courts (Edward Elgar, 2019) 213–30.
- De Visser, Maartje. (2015). 'A Critical Assessment of the Role of the Venice Commission in Processes of Domestic Constitutional Reform,' American Journal of Comparative Law 63: 963–1008.

- Dothan, Shai. 'Three Paradigms of International Judicial Review in The Most Impeded Branch?: Courts and Judicial Review in Theoretical and Comparative Perspectives' (Christoph Bezemek & Yaniv Roznai 2021).
- Dressel, Björn, Raul Sanchez-Urribarri & Alexander Stroh. (2017). 'The Informal Dimension of Judicial Politics: A Relational Perspective,' Annual Review of Law & Society 13: 413–30.
- Dressel, Björn, Raul Sanchez-Urribarri & Alexander Stroh. (2018). 'Courts and Informal Networks,' Special Issue International Political Science Review 39(5).
- Duarte, Madalena, Ana Oliveira, Conceição Gomes & Paula Fernando. (2014). 'The Feminization of the Judiciary in Portugal: Dilemmas and Paradoxes,' Utrecht Law Review 10: 29–43.
- Dunoff, Jeffrey & Pollack, Mark A. (2017). 'The Judicial Trilemma,' American Journal of International Law 111: 225–76.
- Dunoff, Jeffrey L. & Mark A. Pollack. (2018). 'International Judicial Practices: Opening the "Black Box" of International Courts,' Michigan Journal of International Law 40: 47–113.
- Eisenberg, Theodore, Talia Fisher & Issi Rosen-Zvi. (2012). 'Does the Judge Matter? Exploiting Random Assignment on a Court of Last Resort to Assess Judge and Case Selection Effects,' Journal of Empirical Legal Studies 9: 246–90.
- ENCJ, 'Compendium on Councils for the Judiciary' (2021) available at https://pgwrk-websitemedia.s3.eu -west-1.amazonaws.com/production/pwk-web-encj2017-p/Reports/The%20ENCJ%20Compendium%20on%20Councils%20for%20the%20Judiciary%20-%20adopted%20EGA%2029%20October%20Vilnius%20coverpage.pdf.
- Escobar-Lemmon, Maria, Valerie J. Hoekstra, Alice J. Kang & Miki Caul Kittilson. Reimagining the Judiciary: Women's Representation on High Courts Worldwide (Oxford University Press, 2021).
- Esen, Berk & Sebnem Gumuscu. (2016). 'Rising competitive authoritarianism in Turkey,' Third World Ouarterly 37: 1581–1606.
- Fagan, Adam. (2019). 'Reforming judicial recruitment and training in Bosnia-Herzegovina and Serbia under EU guidance: implementation without institutionalisation?,' Journal of European Integration
- Feld, L.P. & S. Voigt. (2003). 'Economic growth and judicial independence: cross-country evidence using a new set of indicators,' European Journal of Political Economy 19: 497–527.
- Garoupa, Nuno & Tom Ginsburg. Judicial Reputation (University of Chicago Press, 2015).
- Gee, Graham, Robert Hanzell, Kate Malleson & Patrick O'Brien. The Politics of Judicial Independence in the UK's Changing Constitution (Cambridge University Press, 2015).
- Geyh, Charles. (2021). 'Judicial Independence at Twilight,' Case Western Reserve Law Review 71: 1045–1119.
- Ginsburg, Tom and Nuno Garoupa. (2009). 'Judicial Audiences and Reputation: Perspectives from Comparative Law,' Columbia Journal of Transnational Law 47: 451.
- Goelzhauser, Greg. (2018). 'Classifying judicial selection institutions,' State Politics & Policy Quarterly
- Goldar, Mónica García. (2020). 'The Glass Ceiling at the Highest Levels of the Spanish Judiciary,' International Journal of the Legal Profession 27: 189–202.
- Grossman, Nienke. (2012). 'Sex on the Bench: Do Women Judges Matter to the Legitimacy of International Courts?,' Chicago Journal of International Law 12: 647–84.
- Grove, Tara Leigh. (2018). 'The Origins (and Fragility) of Judicial Independence,' Vanderbilt Law Review 71: 465–545.
- Guarnieri, Carlo. (2013). 'Judicial Independence in Europe: Threat or Resource for Democracy,' Journal of Representative Democracy 49: 347–59.
- Gutmann, J. & S. Voigt. (2018). 'Judicial Independence in the EU: a puzzle,' European Journal of Law and Economics 49: 83–100.
- Hammergren, Linn A. 'Do Judicial Councils Further Judicial Reform? Lessons from Latin America Carnegie Endowment Rule of Law Series' (2002) Carnegie Endowment Working Papers 28.
- Harper, Erica & Yann Colliou. (2022). 'Re-Imagining Customary Justice Systems: Interrogating Past Assumptions and Entertaining New Ones.' HJRL 11: 1–20.
- Harwa. Ibrahim. 'Nigeria. Women Judges Enhancing the Judiciary' in Gretchen Bauer & Josephine Dawuni (eds) Gender & the Judiciary in Africa: From Obscurity to Parity? (Routledge, 2016) 67–80.

Havelková, Barbara, David Kosař & Marína Urbániková. (2021). 'The Family Friendliness That Wasn't: Access, but Not Progress, for Women in the Czech Judiciary,' Law & Social Inquiry 1-31.

Hayo, B. & S. Voigt. (2016). 'Explaining constitutional change: The case of judicial independence.' International Review of Law & Economics 48: 1–13.

Helmke, Gretchen & Steven Levitsky. (2004). 'Informal institutions & comparative politics: a research agenda,' Perspectives on Politics 2: 725-40.

Helmke, Gretchen & Steven Levitsky. Informal Institutions and Democracy: Lessons from Latin America (The Johns Hopkins University Press, 2006).

Helmke, Gretchen. Institutions on the Edge: The Origins and Consequences of Inter-Branch Crises in Latin America (Cambridge University Press, 2017).

Issacharoff, Samuel & Trevor Morrison. (2020). 'Constitution by Convention,' California Law Review

Iyer, Sundeep. (2013). 'The fleeting benefits of appointments commissions for judicial gender equity.' Commonwealth & Comparative Politics 51: 97–121.

Jakab, András. (2020). 'Informal Institutional Elements as Both Preconditions and Consequences of Effective Formal Legal Rules: The Failure of Constitutional Institution-Building in Hungary.' American Journal of Comparative Law 68: 760-800.

Jasper, Alon. (2021). 'Diversity Through Bureaucracy: System Judges and Intersectional Diversification of the Israeli Judiciary,' The Law & Ethics of Human Rights 15: 313-41.

Jordana, J. & D. Sancho. 'Regulatory Designs, Institutional Constellations and the Study of the Regulatory State' in J. Jordana & D. Levi-Faur. The Politics of Regulation: Institutions and Regulatory Reforms for the Age of Governance (Edward Elgar, 2004) 293-31.

Judith Resnik & Lane Dilg. (2006). 'Responding to a Democratic Deficit: Limiting the Powers and the Term of the Chief Justice of the United States,' U. PA. L. REV. 154: 1575.

Kadlec Ondřej & David Kosař. (2022). 'Romanian version of the rule of law crisis comes to the ECI: The AFJR case is not just about the Cooperation and Verification Mechanism,' Common Market Law

Kadlec, Ondřej, David Kosař & Katarína Šipulová. 'Judicial Councils as Guarantor Institutions -Towards a Post-Partisan Understanding of Judicial Governance' (unpublished manuscript, on file with authors, 2022).

Kalem, Seda. (2020). 'Being a Woman Judge in Turkish Judicial Culture,' International Journal of the Legal Profession 27: 119-44.

Karlsson, Haukur Logi. (2022). 'The Emergence of the Established "By Law" Criterion for Reviewing European Judicial Appointments,' German Law Journal 23: 1051-70.

Kenney, Sally Jane. Gender and Justice: Why Women in the Judiciary Really Matter (Routledge, 2016). Khaitan, Tarun. (2020). 'Killing a Constitution with a Thousand Cuts: Executive Aggrandizement and Party-state Fusion in India,' Law & Ethics of Human Rights 14: S49-S95.

Khaitan, Tarun. (2021). 'Guarantor Institutions,' Asian Journal of Comparative Law 16: S40-S59. Kischel, Uwe. (2012). 'Party, pope, and politics? The election of German Constitutional Court Justices in

comparative perspective,' International Journal of Constitutional Law 11: 962-80. Kochenov, Dimitry & Laurent Pech. Respect for the Rule of Law in the Case Law of the European Court

of Justice: A Casebook Overview of Key Judgments since the Portuguese Judges Case (SIEPS, 2021). Kosař, David & Adam Blisa. (2018). 'Court Presidents: The Missing Piece in the Puzzle of Judicial Governance, German Law Journal 19: 2031–76.

Kosař, David & Samuel Spáč. (2021). 'Post-communist Chief Justices in Slovakia: From Transmission Belts to Semi-autonomous Actors?' Hague Journal on the Rule of Law 1: 107-42.

Kosař, David & Attila Vincze. (2022). 'European Standards of Judicial Governance: From Soft Law Standards to Hard Law,' Journal für Rechtspolitik 30: 491-501.

Kosař, David. (2018). 'Beyond Judicial Councils: Forms, Rationales and Impact of Judicial Self-Governance in Europe,' German Law Journal 19: 1567-1612.

Kosař, David. (2017). 'Politics of Judicial Independence and Judicial Accountability in Czechia: Bargaining in the Shadow of the Law between Court Presidents and the Ministry of Justice' EuConst

Kosař, David. Perils of Judicial Self-Government (Cambridge University Press, 2016).

Kühn Zdeněk The Judiciary in Central and Eastern Europe. Mechanical Jurisprudence in Transformation? (Brill, 2011).

Lauth, H.J. Formal and Informal Institutions (Routledge, 2015).

Lawrence, Sonia. 'Reflections: On judicial diversity and judicial independence' in Adam M. Dodek and Lorne Sossin (eds) Judicial independence in context (Irwin Law, 2010) 193-217.

Ledeneva Alena. (2008). 'Telephone Justice in Russia,' Post-Soviet Affairs 24(4): 324. Lee, HP (ed). Judiciaries in Comparative Perspective (Cambridge University Press, 2011).

Leloup, Mathieu. (2020). 'An Uncertain First Step in the Field of Judicial Self-government,' European Constitutional Law Review 16: 145.

Leloup, Mathieu & David Kosař. (2022). 'Sometimes Even Easy Rule of Law Cases Make Bad Law: Grzeda v Poland,' European Constitutional Law Review (forthcoming).

Levi-Faur, David. (2009). 'Regulatory capitalism and the reassertion of the public interest,' Policy and Society 27: 181-191.

Llanos, Mariana et al. (2015). 'Informal Interference in the Judiciary in New Democracies: A Comparison of Six African and Latin American Cases,' Democratization 23: 1236.

Lurie, G., A. Reichmann & Y. Sagy. (2019). 'Agencification and the administration of courts in Israel,' Regulation and Governance 14: 718-40.

Lurie, Guy. (2021). 'Ministerial Emergency Powers Over Court Administration in the Israeli Judiciary,' International Journal for Court Administration 12: 1-15.

Mak, Elaine. (2008). 'The European Judicial Organisation in a New Paradigm: The Influence of Principles of "New Public Management" on the Organisation of the European Courts,' European Law Journal 14: 718-34.

Malleson, Kate & Peter H. Russell (eds). Appointing Judges in an Age of Judicial Power: Critical Perspectives from Around the World (University of Toronto Press, 2006).

Malleson, Kate. 'The New Judicial Appointments Commission in England and Wales' in Kate Malleson & Peter H. Russell (eds) Appointing Judges in an Age of Judicial Power: Critical Perspectives from Around the World (University of Toronto Press, 2006).

Marcondes, Diego, Cláudia Peixoto & Julio Michael Stern. (2019). 'Assessing randomness in case assignment: the case study of the Brazilian Supreme Court,' Law, Probability and Risk 18: 97-114.

Masengu, Tabeth. (2020). 'The Judicial Service Commission and the Appointment of Women: More to It Than Meets the Eye,' International Journal of the Legal Profession 27: 161-74.

Mathen, Carissima. (2015). 'The Shadow of Absurdity and the Challenge of Easy Cases: Looking Back on the Supreme Court Act Reference,' Supreme Court Law Review 71: 161-89.

Mathieu, E., K. Verhoest & J. Matthys. (2016). 'Measuring multi-level regulatory governance: Organizational proliferation, coordination, and concentration of influence,' Regulation & Governance

Melton, James & Tom Ginsburg. (2014). 'Does de Jure Judicial Independence Really Matter? A Reevaluation of Explanations for Judicial Independence,' Journal of Law and Courts 2: 187-217.

Mikuli, P., N. Fox & R. Puchta (eds). Ministers of Justice in Comparative Perspective (Eleven Publishing, 2019).

Moraru, Madalina & Raluca Bercea. (2022). 'The First Episode in the Romanian Rule of Law Saga: Joined Cases C-83/19, C-127/19, C-195/19, C-291/19, C-355/19 and C-397/19, Asociația Forumul Judecătorilor din România, and their follow-up at the national level,' European Constitutional Law

North, Douglass C. (1991). 'Institutions,' The Journal of Economic Perspectives 5: 97-112.

O'Brien, Patrick. (2018). 'Never let a Crisis go to Waste: Politics, Personality and Judicial Self-Government in Ireland,' German Law Journal 19: 1871-1900.

Olcay, Tarik. (2017). 'Firing Bench-mates: The Human Rights and Rule of Law Implications of the Turkish Constitutional Court's Dismissal of Its Two Members,' EuConst 13:568.

Oxtoby, Chris. 2021. 'The Appointment of Judges: Reflections on the Performance of the South African Judicial Service Commission,' Journal of Asian and African Studies 56: 34-47.

Özbudun, Ergun. (2015). 'Turkey's Judiciary and the Drift Toward Competitive Authoritarianism,' The International Spectator 50: 42–55.

Parau, Cristina E. Transnational Networking and Elite Self-empowerment (Cambridge University Press, 2018).

- Peresie, Jennifer L. (2005). 'Female Judges Matter: Gender and Collegial Decisionmaking in the Federal Appellate Courtsm' The Yale Law Journal 114: 1759–1790.
- Petrov, Jan. (2021). '(De-)judicialization of politics in the era of populism: lessons from Central and Eastern Europe,' The International Journal of Human Rights 26: 1181–1206.
- Pierson, Paul. (2000). 'Increasing Returns, Path Dependence, and the Study of Politics,' American Pol. Science Review 94: 251–67.
- Popova, Maria & Beers, Daniel. (2020). 'No Revolution of Dignity for Ukraine's Judges: Judicial Reform after the Euromaidan,' Democratizatsiya 28: 113–42.
- Popova, Maria. (2012b). 'Why Doesn't the Bulgarian Judiciary Prosecute Corruption?,' Problems of Post-Communism 59: 35–49.
- Popova, Maria. Politicized Justice in Emerging Democracies: A Study of Courts in Russia and Ukraine (Cambridge University Press, 2012a).
- Pozas-Loyo, Andrea & Julio Rios-Figueroa. (2018). 'Anatomy of an informal institution: The "Gentlemen's Pact" and judicial selection in Mexico, 1917–1994, International Political Science Review 39: 647–61.
- Pozas-Loyo, Andrea & Julio Rios-Figueroa. (2023). 'Instituciones informales e independencia judicial de facto. El eslabón olvidado en el camino hacia la eficacia institucional,' Política y gobierno 29: 1–27.
- Rackley, Erika. Women, Judging and the Judiciary (Routledge, 2013).
- Randazzo, Kirk A. (2008). 'Strategic Anticipation and the Hierarchy of Justice in U.S. District Courts,' American Politics Research 36: 669–93.
- Remembering Benedicto Kiwanuka. The Judiciary Insider No. 11/2018.
- Resnik, Judith. (2021). 'Representing What? Gender, Race, Class and the Struggle for the Identity and the Legitimacy of Courts,' The Law & Ethics of Human Rights 15: 1–91.
- Sadurski, Wojciech. Poland's Constitutional Breakdown (Oxford University Press, 2019).
- Sallusti, Alessandro and Luca Palamara. Il Sistema: Potere, politica affari: storia segreta della magistratura (Rizzoli, 2021).
- Schultz, Ulrike, & Gisela Shaw. 'Introduction: Gender and Judging: Overview and Synthesis' in Ulrike Schultz & Gisela Shaw (eds) Gender and Judging (Hart Publishing, 2013) 1–50.
- Selejan-Gutan, Bianca. (2018). 'Romania: Perils of a "Perfect Euro-Model" of Judicial Council,' German Law Journal 19: 1707–40.
- Sengupta, Arghya & Ritwika Sharma. Appointment of Judges to the Supreme Court of India: Transparency, Accountability, and Independence (Oxford University Press, 2018).
- Šipulová, Katarína, Samuel Spáč, David Kosař, Tereza Papoušková & Viktor Derka. Judicial Self-Governance Index: Towards Better Understanding of the Role of Judges in Governing the Judiciary (Regulation and Governance, 2022).
- Sirota, Léonid. (2011). 'Towards a Jurisprudence of Constitutional Conventions,' OUCLJ 11: 29–51.
- Śledzińska-Simon, Anna. (2018). 'The Rise and Fall of Judicial Self-Government in Poland: On Judicial Reform Reversing Democratic Transition,' German Law Journal 19: 1839–70.
- Smithey, S. and J. Ishiyama. (2002). 'Judicious choices: designing courts in post-communist politics,' Communist and Post-Communist Studies 33: 163–82.
- Sofos, Spyros. (2020). 'Charting the waters: populism as a gendered phenomenon,' Open Democracy. 27 January 2020.
- Songer Donald R. & Susan W. Johnson. (2007). 'Judicial Decision Making in the Supreme Court of Canada: Updating the Personal Attribute Model,' Canadian Journal of Political Science/Revue canadienne de science politique 40: 911–34.
- Sonnevold, Nadia. 'Women's Access to Legal Education and Their Appointment to the Judiciary: The Dutch, Egyptian, and Indonesian Cases Compared' in Nadia Sonnevold & Monika Lindbekk, Women Judges in the Muslim World. A Comparative Study of Discourse and Practice (Brill, 2017).
- Spáč, Samuel. (2018). 'Recruiting European Judges in the Age of Judicial Self-Government,' German Law Journal 19: 2077–2104.
- Spáč, Samuel. (2020). 'The Illusion of Merit-Based Judicial Selection in Post-Communist Judiciary: Evidence from Slovakia,' Problems of Post-Communism 1–11.
- Spáč, Samuel, Katarína Šipulová & Marína Urbániková. (2018). 'Capturing the Judiciary from Inside: The Story of Judicial Self-Governance in Slovakia,' German Law Journal 19: 1741–68.

- Stephenson, Scott. (2021). 'Constitutional Conventions and the Judiciary' OJLS 41: 750-75.
- Tate, C. Neal & Roger Handberg. (1991). 'Time Binding and Theory Building in Personal Attribute Models of Supreme Court Voting Behavior, 1916–88,' American Journal of Political Science 35: 460–480.
- Taylor, Greg. (2014). 'Convention by consensus: Constitutional conventions in Germany,' ICON 12: 303-29.
- Taylor, Matthew. (2014). 'The Limits of Judicial Independence: A Model with Illustration from Venezuela under Chavez,' Journal of Latin America Studies 46: 229.
- Toraasen, Marianne. (2022). 'Women's Judicial Representation in Haiti: Unintended Gains of State-Building Efforts,' Politics&Gender 1–32.
- Torres Pérez. (2018). 'Judicial Self-Government and Judicial Independence: the Political Capture of the General Council of the Judiciary in Spain,' German Law Journal 19: 1769–1800.
- Tsereteli, Nino. (2022). 'Backsliding into Judicial Oligarchy? The Cautionary Tale of Georgia's Failed Judicial Reforms, Informal Judicial Networks and Limited Access to Leadership Positions,' Review of Central and East European Law 47: 167–201.
- Tsereteli, Nino. (2020). 'Judicial recruitment in post-communist context: informal dynamics and façade reforms,' International Journal of the Legal Profession, 10.1080/09695958.2020.1776128.
- Tushnet, Mark. The New Fourth Branch (Cambridge University Press, 2021).
- Uitz, Renata. (2015). 'Can you tell when an illiberal democracy is in the making? An appeal to comparative constitutional scholarship from Hungary,' ICON 13: 279–300.
- Valdini, Melody E. & Christopher Shortell. (2016). 'Women's Representation in the Highest Court: A Comparative Analysis of the Appointment of Female Justices,' Political Research Quarterly 69: 865–76.
- Van Dijk, Frank. Perceptions of the Independence of Judges in Europe: Congruence of Society and Judiciary. (Palgrave Macmillan, 2021).
- Varol, Ozan O., Lucia D. Pellegrina & Nuno Garoupa. (2017). 'An Empirical Analysis of Judicial Transformation in Turkey,' Am. J. Comp. L. 65: 187–216.
- Vasek, Markus. Richterbestellung in Österreich (Verlag Österreich, 2022).
- Vauchez, Antoine. (2018). 'The Strange Non-Death of Statism: Tracing the Ever Protracted Rise of Judicial Self-Government in France,' German Law Journal 19: 1613–40.
- Verhoest, K. 'Agencification Processes and Agency Governance: Organizational Innovation at a Global Scale?' in P. Valkama, S.J. Bailey & A.-V. Anttiroiko. Organizational Innovation in Public Services: Forms and Governance (Palgrave Macmillan, 2013) 49–71.
- Volcansek, Mary. (2009). 'Exporting the Missouri Plan: Judicial Appointment Commissions,' Missouri Law Review 74: 783–800.
- Wang, Juan & Liu, Sida. (2021). 'Institutional proximity and judicial corruption: A spatial approach,' Governance 35: 633–49.
- Weinshall-Margel, Keren. (2011). 'Attitudinal and Neo-Institutional Models of Supreme Court Decision Making: An Empirical and Comparative Perspective from Israel,' Journal of Empirical Legal Studies 8: 556–86.
- Weiler, Joseph H.H. (2023) 'Cry, the Beloved Country', Verfassungsblog, 1 February 2023, https://verfassungsblog.de/cry-beloved-country/.
- Weinshall, Keren. (2022). 'Courts and Diversity: Normative Justifications and Their Empirical Implications,' The Law and Ethics of Human Rights 15: 187–220.
- White, Brent T. (2009). 'Rotten to the Core: Project Capture and the Failure of Judicial Reform in Mongolia,' East Asia Law Reform 4: 209.
- Williams, Margaret S. & Frank C. Thames. (2008). 'Women's Representation on High Courts in Advanced Industrialized Countries,' Politics & Gender 4: 451–71.
- Wittreck, Fabian. (2018). 'German Judicial Self-Government Institutions and Constraints,' German Law Journal 19: 1931–50.
- Zgut, Edit. (2022). 'Informal Exercise of Power: Undermining Democracy Under the EU's Radar in Hungary and Poland,' HJRL.
- Zheng, Chunyan, Jiahui Ai & Sida Liu. (2017). 'The Elastic Ceiling: Gender and Professional Career in Chinese Courts,' Law & Society Review 51: 168–99.